Author's response to reviews

Title: Residents' views about family medicine specialty education in Turkey

Authors:

Arzu Uzuner (arzuuzuner@gmail.com)
Pinar Topsever (topsever@gmail.com)
Ilhami Unluoglu (iunluog@ogu.edu.tr)
Ayse Caylan (acaylan2000@yahoo.com)
Nezih Dagdeviren (dagdeviren@trakya.edu.tr)
Yesim Uncu (yesimuncu@uludag.edu.tr)
Mumtaz Mazicioglu (mazici@erciyes.edu.tr)
Alis Ozcakir (alis@uludag.edu.tr)
Hakan Ozdemir (hozdemir@uludag.edu.tr)
Fusun Ersoy (ersoyfuture@gmail.com)

Version: 4 Date: 2 March 2010

Author's response to reviews: see over
To Anastasios Koutsos, PhD.

Scientific Editor
BMC-series Journals
BioMed Central

“RESIDENTS’ VIEWS ABOUT FAMILY MEDICINE SPECIALTY EDUCATION IN TURKEY”
MS: 4844899643155718

According to the editorial advices we have sent the above manuscript to a native English speaker Professor Ray W. Guillery (rguiller@facstaff.wisc.edu). He kindly revised the manuscript and we added an acknowledgement for him in the text.

If there are still language problem, we will apply for additional professional help to the sources indicated in your mail (Manuscript Presentation Service (www.biomedes.co.uk), International Science Editing (http://www.internationalscienceediting.com/) and English Manager Science Editing (http://www.sciencemanager.com/).

We also revised the manuscript format according to the journal style. The abstract is shortened to 350 words and also the references were revised.

Many thanks for the editorial support.

With my best regards

Arzu Uzuner
Marmara University Medical School
Department of Family Medicine
Istanbul/Turkey